Author's response to reviews

Title: Antidiabetic and renoprotective effects of the chloroform extract of Terminalia chebula Retz. seeds in streptozotocin-induced diabetic rats

Authors:

Nalamolu Koteswara Rao (nalamolu_k@rediffmail.com)
Srinivas Nammi (nammi@rediffmail.com)

Version: 5 Date: 15 April 2006

Author's response to reviews: see over
Format Check

**Title:** Antidiabetic and renoprotective effects of the chloroform extract of *Terminalia chebula* Retz. seeds in streptozotocin-induced diabetic rats

Manuscript No: 1787908154687251

Journal: BMC Complementary and Alternative Medicine

The manuscript has been checked for the format of BMC Complementary and Alternative Medicine and the following sections are added to the revised manuscript as suggested.

**Competing interests**

The authors declare that they have no competing interests.

**Authors’ contributions**

NKR conceived the study, made significant contributions in data analysis, data interpretation, writing of the manuscript and in coordination of the experiments. SN made substantial contributions in conceptualization of statistical analyses, drafting the final manuscript and designing the illustrations. All authors read and approved the final manuscript.

**Acknowledgements**

The authors are grateful to Sun Pharmaceuticals Limited (Baroda, India) for supplying glibenclamide as a generous gift sample. The authors are also thankful to the Department of Pharmaceutical Sciences Andhra University, Visakhapatnam, India for providing all the facilities to carryout the studies.